Title: A case of delayed neurological manifestation following Carbon Monoxide poisoning in Sri Lanka: Epidemiology of exposure and literature review

Authors:

Prabhashini Kumarihamy (prabhashinikumarihamy@yahoo.com)
Senanayake Abeysinghe Mudiyanelage Kularatne (tbn1917@gmail.com)
Manoji Pathirage (manojipathi@yahoo.com)
Wasala Mudiyanelage Sujeewa Nilanthi Gunaratne (sujeewa.gunaratne@gmail.com)
Roshitha Waduge (roshitha2006@yahoo.com)

Version: 3 Date: 25 Mar 2019

Author’s response to reviews:

Dr. Prabhashini Kumarihamy.
Senior Registrar in Medicine,
Professorial Medical Unit,
Teaching Hospital Peradeniya.
25/03/2019
The Editor,
BMC Pharmacology and Toxicology

Dear sir/madam.

CASE REPORT : A case of delayed neurological manifestation following Carbon Monoxide poisoning in Sri Lanka: Epidemiology of exposure and literature review

Thank you for considering our manuscripts. I am sorry that I was unable to submit the revised manuscript on time. I have done all the requested changes and answered to all question for editor and reviewers’ comments. I have attached the clean version of revised manuscript. Details of the changers and answers to editor and reviewers’ comments are mention below in this letter.

I would be grateful to you if you could consider this article to published in your journal. Please contact me if you need further correction and I am happy to reply them.
Best regards,

Prabhashini Kumarihamy.

Point by point answer

Technical Comments:

1. Figure files should contain only the image, as well as any associated keys/annotations. If legends are present within the figure files, please remove them.

Legends were removed from figure files

2. Please provide figure legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript or the figure files, please move them.

Figure legends were included under ‘figure legends’ after the references.